Author’s response to reviews

Title: The study protocol of a cluster-randomised controlled trial of family-mediated personalised activities for nursing home residents with dementia

Authors:

Eva S van der Ploeg (Eva.vanderPloeg@monash.edu)
Cameron J Camp (Cameron@cen4ard.com)
Barbara Eppingstall (Barbara.Eppingstall@monash.edu)
Susannah J Runci (Susannah.Runci@monash.edu)
Daniel W O’Connor (Daniel.OConnor@monash.edu)

Version: 2 Date: 7 December 2011

Author’s response to reviews:

To: Biomed Geriatrics

December 7th 2011

Dear sir/madam,

Please find attached the revised manuscript named:

‘The study protocol of a cluster-randomised controlled trial of family-mediated personalised activities for nursing home residents with dementia’

for publication in Biomed Geriatrics. This manuscript is neither under review at any other journal, nor has it been published elsewhere before. The authors declare that they have no competing interests. We hope that this revised version of the study protocol is acceptable for publication.

Sincerely yours,

Dr. Eva van der Ploeg